Author’s response to reviews

Title: Effects of simultaneous cognitive and aerobic exercise training on dual-task walking performance in healthy older adults: Results from a pilot randomized controlled trial

Authors:

David Raichlen (draichle@usc.edu)
Pradyumna Bharadwaj (prad@emaiol.arizona.edu)
Lauren Nguyen (languyen@email.arizona.edu)
Mary Kathryn Franchetti (mkfranchetti@email.arizona.edu)
Erika Zigman (erikazigman@live.com)
Abigail Solorio (abbysolorio@email.arizona.edu)
Gene Alexander (gea@email.arizona.edu)

Version: 3 Date: 17 Feb 2020

Author’s response to reviews:

Editor Comments:

1. In line with journal requirements, please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.

We have added a conclusions section after the discussion section.

2. Currently, the majority of figures/tables in the supplementary materials are not referred to in the text. Please ensure all figures/tables are present and referred to in the text.

We have now referred to all supplementary tables and figures in the text.

3. Please remove the CONSORT checklist from the manuscript, as at this stage it is no longer necessary.

We have removed the checklist.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should
remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

We have uploaded a clean manuscript file.